Author’s response to reviews

Title: Conceptualising a model to guide nursing and midwifery in the community guided by an evidence review

Authors:

Patricia Leahy-Warren (patricia.leahy@ucc.ie)
Helen Mulcahy (helen.mulcahy@ucc.ie)
Lazelle Benefield (lazellebenefield@ucc.ie)
Colin Bradley (C.Bradley@ucc.ie)
Alice Coffey (a.coffey@ucc.ie)
Ann Donohoe (ann.donohoe@ucd.ie)
Serena Fitzgerald (serena.fitzgerald@ucc.ie)
Tim Frawley (timothy.frawley@ucd.ie)
Elizabeth Healy (elizabeth.healy@hse.ie)
Maria Healy (maria.healy@ucd.ie)
Marcella Kelly (marcella.kelly@nuigalway.ie)
Bernard McCarthy (bernard.mccarthy@nuigalway.ie)
Kathleen McLoughlin (kemcloughlin@gmail.com)
Catherine Meagher (catherine.meagher@nuigalway.ie)
Rhona O'Connell (r.oconnell@ucc.ie)
Aoife O'Mahony (110316715@umail.ucc.ie)
Gillian Paul (gillian.paul@ucd.ie)
Amanda Phelan (amanda.phelan@ucd.ie)
Diarmaid Stokes (diarmuid.stokes@ucd.ie)
Jessica Walsh (jessica.lower@ucd.ie)
Dear Editor,

please find below responses to comments:

Editor comment: The abbreviation DHPN - the explanation of this abbreviation should just show what the letters stand for. Please amend the text in the brackets on page 6 to show what DHPN stands for.

Response: This has been revised as suggested.

Editor comment: Page 16 line 382 a typo has been introduced here. Change nurses to nurse.

Response: this has been changed as requested

Editor comment: Remove NNU from the abbreviation list as it is now spelled out.

Response: this has been removed as requested

Kind regards

Patricia